Author's response to reviews

Title: Measuring walking within and outside the neighborhood in Chinese elders: reliability and validity

Authors:

Ester Cerin (ecerin@hku.hk)
Anthony Barnett (abarnett@hku.hk)
Cindy HP Sit (sithp@hku.hk)
Martin Cheung (martin_cheung@dh.gov.hk)
Janet Lee (h0460544@hku.hk)
Daniel SY Ho (syho@hku.hk)
Wai-man Chan (w_m_chan@dh.gov.hk)

Version: 5 Date: 30 October 2011

Author's response to reviews: see over
We thank Reviewer 1 and the Associate Editor for their interest in our manuscript.

We have now addressed the remaining comment raised by Dr. Sharpe pertaining to the need to provide a citation for a paper that we refer to in the Discussion section and has been now accepted for publication in the Journal of Aging and Physical Activity.

Response to comment by Reviewer 1:

Reviewer 1: Dr. Patricia Sharpe

The authors have made the requested revisions. One issue remains uncertain: The authors want to discuss the findings from another paper that is not yet in print. They believe the other paper will be accepted for publication soon. If that is the case, then that paper needs to be cited in this one. Otherwise the discussion section is less useful to the reader. This paper should be able to stand alone

Our reply:

We thank Dr. Sharpe for her positive appraisal of our manuscript. We have now added a citation for the paper mentioned above: